

Make Your Publications Visible.

A Service of



Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre

Metwally, Abdelmoneim Bahyeldin Mohamed; Elsharkawy, Ahmed Abdelaty M.; Salem, Mohamed Ibrahim

Article

The impact of corporate social responsibility on operating cash flow opacity: the moderating role of tax avoidance

Cogent Business & Management

Provided in Cooperation with:

Taylor & Francis Group

Suggested Citation: Metwally, Abdelmoneim Bahyeldin Mohamed; Elsharkawy, Ahmed Abdelaty M.; Salem, Mohamed Ibrahim (2024): The impact of corporate social responsibility on operating cash flow opacity: the moderating role of tax avoidance, Cogent Business & Management, ISSN 2331-1975, Taylor & Francis, Abingdon, Vol. 11, Iss. 1, pp. 1-20, https://doi.org/10.1080/23311975.2024.2390692

This Version is available at: https://hdl.handle.net/10419/326487

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.



https://creativecommons.org/licenses/by/4.0/

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.





Cogent Business & Management



ISSN: 2331-1975 (Online) Journal homepage: www.tandfonline.com/journals/oabm20

The impact of corporate social responsibility on operating cash flow opacity: the moderating role of tax avoidance

Abdelmoneim Bahyeldin Mohamed Metwally, Ahmed Abdelaty M. Elsharkawy & Mohamed Ibrahim Salem

To cite this article: Abdelmoneim Bahyeldin Mohamed Metwally, Ahmed Abdelaty M. Elsharkawy & Mohamed Ibrahim Salem (2024) The impact of corporate social responsibility on operating cash flow opacity: the moderating role of tax avoidance, Cogent Business & Management, 11:1, 2390692, DOI: 10.1080/23311975.2024.2390692

To link to this article: https://doi.org/10.1080/23311975.2024.2390692

9	© 2024 The Author(s). Published by Informa UK Limited, trading as Taylor & Francis Group
	Published online: 16 Aug 2024.
	Submit your article to this journal ${\it \mathbb{G}}$
ılıl	Article views: 3166
Q ^L	View related articles ☑
CrossMark	View Crossmark data ☑
4	Citing articles: 10 View citing articles ☑



ACCOUNTING, CORPORATE GOVERNANCE & BUSINESS ETHICS | RESEARCH ARTICLE



The impact of corporate social responsibility on operating cash flow opacity: the moderating role of tax avoidance

Abdelmoneim Bahyeldin Mohamed Metwally^{a,b} (i), Ahmed Abdelaty M. Elsharkawy^c (i) and Mohamed Ibrahim Salem^d (i)

^aDepartment of Accounting, College of Business Administration, King Faisal University, Saudi Arabia; ^bDepartment of Accounting, Faculty of Commerce, Assiut University, Assiut, Egypt; 'Department of Accounting, Faculty of Business, Ain Shams University, Cairo, Egypt; dDepartment of Accounting, Faculty of Commerce, Suez University, Suez, Egypt

ABSTRACT

This study examines the association between corporate social responsibility (CSR) and operating cash flow opacity (OCFOP). The current study also aims to examine the moderating effect of corporate tax avoidance on the relationship between CSR and OCFOP. The sample used in this research incorporates non-financial companies listed on the EGX from 2012 to 2021. The final sample comprised of 52 companies with 520 observations. Statistical analysis was performed using pooled OLS and random effects regression analysis. The current research indicates that there is a significant positive relationship between companies engaging in CSR practices and having a higher level of OCFOP. Further, corporate tax avoidance is found to have a positive moderating role as it strengthens the relationship between CSR and OCFOP. The current research findings have several implications for Egyptian and emerging market regulators, investors and shareholders in the Egyptian market, including the fact that CSR practices are not always an indicator of ethical behavior, as they may be used to conceal unethicality underneath. To the best of our knowledge, this study represents an initial endeavor to elucidate the moderating impact of corporate tax avoidance on the correlation between CSR and OCFOP in an emerging market context. This contribution expands upon the existing literature on CSR and opacity by presenting novel empirical findings from an emerging market, thereby illuminating the moderating function of tax avoidance. Early investigations primarily concentrated on the direct link between CSR and OCFOP.

ARTICLE HISTORY

Received 29 February Revised 30 July 2024 Accepted 6 August 2024

KEYWORDS

Operating cash flows opacity; corporate social responsibility; tax avoidance: emerging economy; Egypt

SURIFCTS

Business, Management and Accounting; Cultural Studies; African Studies

1. Introduction

Corporate Social Responsibility (CSR) constitutes a globally pertinent subject of scholarly investigation, particularly within enterprises that exhibit a proclivity for comprehensive reporting of their CSR endeavors (Alatawi et al., 2023). The emphasis lies in scrutinizing the efficacy with which these entities fulfil their dedication to furnishing candid revelations pertaining to both pecuniary and non-monetary facets (Jouber, 2019a). Previous research in less developed countries (LDCs) connected CSR practices to legitimizing corporate actions (El-Bassiouny & Letmathe, 2018), earning quality (Bozzolan et al., 2015; Mohmed et al., 2019), earning management (Abdelfattah & Elfeky, 2021; Liu & Lee, 2019; Sial et al., 2018), firm value (Aboud & Diab, 2018), firm performance (Elmghaamez et al., 2023; Sial et al., 2018), employees creativity (Abdelmotaleb et al., 2018), social and political problems that led to different practices (Diab & Metwally, 2020), cost reduction as it was found that CSR disclosure is used as a rhetoric to cover cost reduction (Metwally et al., 2022), firm opacity (Agnes Cheng et al., 2020; Kim & Yoo, 2022; You & Chen, 2022) and tax incentive and avoidance (Du & Li, 2023; Kacem & Brahim Omri, 2021; Van & Ly, 2021).

The diverse themes in studying CSR stem from the aim to link its practices to ethics and optimal approaches. CSR practices serve as markers for companies that broadly adopt ethical and eco-conscious strategies while fostering sustainable community development (Alnaim & Metwally, 2024; Buhaya & Metwally, 2024; Prior et al., 2008; Rakia et al., 2023). However, opportunistic behavior managers may attempt to hide their poor transparency in financial reporting by disclosing their CSR practices (Abdelfattah & Elfeky, 2021; Rahman & Zheng, 2022).

This article aims to extend CSR, opacity and tax avoidance literature (Abdelfattah & Aboud, 2020; Abid & Dammak, 2021; Agnes Cheng et al., 2020; Chouaibi et al., 2021; Du & Li, 2023; Gavious et al., 2022; Hoi et al., 2013; Jiang et al., 2022; Kacem & Brahim Omri, 2021; Kim & Yoo, 2022; Liu & Lee, 2019; Rakia et al., 2023; You & Chen, 2022) through shedding light on a blind spot in the CSR and opacity literature. Thus, corporate tax avoidance has a moderating effect on the relationship between CSR and Operating Cash Flow Opacity (OCFOP). Early studies concentrated on the firm opacity theory and the relationship between CSR and opacity, the relationship between CSR and tax avoidance (Abdelfattah & Aboud, 2020; Abid & Dammak, 2021; Chouaibi et al., 2021; Du & Li, 2023; Gavious et al., 2022; Hoi et al., 2013; Jiang et al., 2022; Kacem & Brahim Omri, 2021; Liu & Lee, 2019; Rakia et al., 2023), and the relationship between tax avoidance and opacity (Arman & Mira, 2021; Lastiati, 2020; Lee & Bose, 2021; Sá & Alves, 2022). Despite an extensive search, no prior studies have specifically addressed the moderating influence of tax avoidance on the relationship between CSR and OCFOP. This notable absence underscores a significant research gap in the literature, particularly concerning the nuanced role of tax avoidance in shaping OCFOP levels within the context of CSR practices.

The Egyptian market is an intriguing area for exploration because of its distinctive features. These distinctive attributes encompass a series of regulatory and structural transformations that can impact CSR on OCFOP and the moderating effect. Since the 1990s, Egypt has undergone a significant shift in its economic framework, transitioning from a centrally planned model under government control to embracing economic reform initiatives, privatization and open-market principles (Ali et al., 2024; Mohamed et al., 2019). This period also witnessed developments in accounting systems aimed at augmenting the quality of financial reporting within the stock exchange in the late 1990s (Metwally, 2022; Metwally et al., 2021). However, it is important to note that, in contrast to more developed markets such as the United States and the UK, as well as the expectations associated with recent development efforts, the Egyptian market still grapples with government interventions and constraints that impede the full adoption of international standards (Abdelazim et al., 2023; Diab et al., 2023; Diab & Metwally, 2020).

Egypt is characterized by a low regulatory enforcement regime, and its accounting systems are heavily influenced by political factors (Abdelazim et al., 2023; Ball et al., 2000; Metwally, 2022). Consequently, this environment often leads to deficiencies in disclosure and effective governance mechanisms, a situation akin to many other emerging markets (Bremer & Elias, 2007; Desoky & Mousa, 2012; Metwally, 2022; Ntim & Soobaroyen, 2013; Rahman & Zheng, 2022). Having said this, it is imperative to investigate CSR and OCFOP within the unique institutional context of Egypt, which differs from the well-established and stable markets. This study aims to determine whether recent economic, governance, regulatory and accounting advancements have been sufficient to improve the OCFOP in Egypt. That is, the current study aims to answer two allied questions to realize the moderating impact of tax avoidance in the Egyptian context: (1) What is the influence of CSR on operating cash flow opacity? (2) Does TA moderate the association between CSR and OCFOP? Our results reveal a positive significant relationship between companies engaging in CSR practices and having a higher level of OCFOP. Further, corporate tax avoidance is found to have a positive moderating role as it strengthens the effect of CSR on OCFOP. Consequently, this study offers evidence that tax avoidance is one of the factors that affect OCFOP.

This study contributes to the literature in various ways. First, most of the earlier research was conducted in developed countries, which are different from the business environment in developing countries such as Egypt. In addition, Egypt is the leading emerging market in Africa, one of the most important financial markets in the Middle East and North Africa (MENA) region (Metwally, 2022; Metwally et al., 2021), sharing unique settings with some countries in the Middle East and North Africa (MENA region, especially after the 2011 and 2013 liberal revolution and Arab spring revolutions (Abdelazim et al., 2023)). Second, we examine the moderating effect of corporate tax avoidance on the relationship between CSR practices and OCFOP in this emerging market. Third, the current research aims to provide several

recommendations and implications from the results to potential investors, shareholders and regulators in emerging markets that share similar economic, political and social aspects as in Egypt. Finally, this study distinguishes itself from earlier research by employing opacity and agency theories as robust theoretical frameworks for the formulation of testable hypotheses and the elucidation of the study's outcomes. These findings contribute to the refutation of assertions made in prior studies that primarily focused on stewardship and stakeholder theories.

The remainder of this article is organized as follows. Section 2 presents the context of the Egyptian Exchange (EGX). Section 3 explains the theoretical framework deployed in the current study. Section 4 provides a detailed literature review and hypothesis development. Section 5 illustrates the current research methodology and methods used to clarify the findings. Section 6 presents the empirical findings and a discussion. Section 7 presents additional analysis through adding additional controls to the analysis. Section 8 illustrates additional measures for robustness analysis and additional tests for endogeneity problem. Finally, section 9 provides concluding remarks, limitations and opportunities for future research.

2. EGX context

The third-most populated nation in Africa is Egypt, which has the highest population density in the Arab world, with a population of 109 million, according to the World Bank (2021). Its economy is diversified, and it is a lower-middle-income nation known to be one of the leading MENA emerging markets. The Egyptian Stock Exchange (EGX) was the primary stock exchange founded in the MENA region and is one of the first-born stock exchanges worldwide (Metwally, 2022; Metwally et al., 2021). Egypt has transitioned into a market-oriented economy over the past two decades, and it has realized that it needs legislative reforms to go along with its economic transformation.

To increase market confidence, Egyptian authorities starting early 90s have collaborated with relevant international organizations by implementing best practices and global standards that support sustainable markets by moving toward a free market and implementing privatization courses to stimulate economic development (Abdelazim et al., 2023; Diab et al., 2023). Moreover, international organizations have successfully advanced several activities in Egypt, and government authorities have published international accounting standards to encourage foreign investment, enhance trustworthiness and ensure the transparent disclosure of financial reports (Abdelfattah & Aboud, 2020; El-Bassiouny & Letmathe, 2018).

The Egyptian economy gradually improved because of a new transformational restructuring agenda launched in 2014 and assistance from some Gulf nations, with the annual growth rate of gross domestic product (GDP) reaching 4.3% in 2015/2016 (UNCTAD, 2017). Also, in 2016 the International Monetary Fund authorized a \$12 billion credit package for three years in November of the same year intending to boost investor confidence and the economy. Egypt received \$8.1 billion more in foreign direct investment in 2017 than in 2016 (World Bank, 2017). In addition, the Egyptian government restructured its taxation system by issuing Law 91 for 2005, which aims to change the taxation structure by accompanying a self-assessment routine for corporate taxes that supposes an increase in tax revenues. In addition to the encouragement of the economic reform program supported by the International Monetary Fund through the introduction of VAT in 2016, it helped Egyptian authorities increase collection for a higher level of taxable revenues, but because there is still a low degree of tax conformity contributing toward aggressive tax practices that result in loss of tax collections than predicted (Abdel-Mowla, 2012).

Unlike the social, political and economic background of emerging financial markets, rather than developed financial markets, this creates special considerations of CSR practices that require further research. In Egypt, stakeholder information strategy is mostly used to share CSR efforts. As part of this strategy, businesses use one-way communication mechanisms to forbid stakeholder participation or input regarding their CSR operations. However, Egypt was the first Arab nation to address sustainability issues and the ESG index. The appearance of the ESG index presents data and non-financial disclosures motivating the CSR practices of the top 100 companies. The index aims to disclose companies' environmental, social and governance practices, and it releases only the top 30 companies annually.

Moreover, CSR practices varied widely before and after the 2011 and 2013 movements, and the pre-2011 and 2013 movement CSR practices were led by private companies and aimed at being committed to enhance employees' local communities and society (Diab & Metwally, 2020). After the 2011 and 2013 movements according to Darrag and Crowther (2017) CSR practices were found to shift toward more socially accountable practices and dues toward society in response to the socio-economic and political changes that occurred in Egypt during this period. Accordingly, our research uses observations and data extracted from 52 non-financial Egyptian companies listed in the ESG index covering 2012 to 2021.

3. Theoretical framework

The impact of CSR on OCFOP was connected mainly to agency theory on most of the studies that reported positive relation between CSR and OCFOP. However, many other theories and perspectives were deployed to explain this relationship. These theories include institutional theory and organizational factors (Brahem et al., 2022; Sial et al., 2018), firm opacity theory (Agnes Cheng et al., 2020; Ball et al., 2000; Hamrouni et al., 2021; Prior et al., 2008; You & Chen, 2022), signaling theory (You & Chen, 2022), stakeholder theory and stewardship theory (Mohmed et al., 2019; Nair et al., 2019). These different theories and perspectives in the literature were deployed to explain different aspects of the relationship between CSR and OCFOP. As some of them were used to explain and support the positive relationship while others were used to explain the negative relationship.

Agency theory explains that the separation between ownership and management was detected with the emergence of big corporations, and therefore, leads to the so-called agency problems (Fama & Jensen, 1983). Agency problems are considered as a cost because of the information asymmetry. Kılıç and Kuzey (2018) explained that the management (agent) is having all the access to the information available, so what would guarantee that the management will utilize the whole information set in support of the owners' (stockholders) benefits instead of his/her remunerations. While opacity causes poor transparency in financial reporting, therefore increasing the costs of the agency and raising the opportunistic behaviour of managers. As a result of opportunistic behaviour by managers, they may use their engagement in CSR practicing presenting a good image of the company's reporting practice thus hiding their true company's performance at this moment CSR would act to provide a diversion tactic rather than aiming to present an initiative for further voluntary disclosure and subsequently reducing the agency costs (Abdelfattah & Elfeky, 2021; Jouber, 2019b; Mohmed et al., 2019; Rahman & Zheng, 2022).

On the contrary, studies that reported negative relationship between CSR and OCFOP deployed mainly two theories stakeholder theory and stewardship theory. Stakeholder theory views the most essential task for a firm as to carry out the stakeholder expectations, and therefore, meet their interest toward the most efficient use of the firm's resources (Grougiou et al., 2019). Based on the theory's concepts it would be expected that companies highly engaged with CSR activities would be less corporate tax avoidance and imply less aggressive accounting which limits the opacity of financial reporting and would enhance financial transparency and reporting quality by employing having fewer earnings opacity (Nair et al., 2019). Besides stewardship theory promotes that corporate managers perform as the optimal agents for their shareholders and therefore would likely maximize their interest in better-making shareholders' economic decisions (Mohmed et al., 2019).

Theoretically, tax avoidance moderating role can be explained through stakeholder theory, legitimacy theory and agency theory. Stakeholders and legitimacy theories represent an opposing view to how agency theory explain the tax avoidance impact on the relationship between CSR and OCFOP. Stakeholder and legitimacy theories explain how companies that promotes CSR practices are expected to be more transparent and have less OCFOP. Hence, firms practicing CSR are willing to maintain social prosperity, and therefore, tax avoidance would show management's act of negligent behaviour toward its firm reputation and stakeholders' interest (Chouaibi et al., 2021; Du & Li, 2023; Hoi et al., 2013; Liu & Lee, 2019; Van & Ly, 2021). On the contrary, agency theory through concentrating on opportunistic behaviour managers is used to explain how managers will tend to increase the CSR practices and disclosure to hide underneath the tax savings through tax avoidances (Gavious et al., 2022). Therefore, CSR practices will induce managers to be tax incentivized resulting in being less transparent and presenting more opacity in disclosing their financial reports (Abdelfattah & Aboud, 2020; Abid & Dammak, 2021; Jiang et al., 2022; Lee & Bose, 2021; Marwat et al., 2021; Zeng, 2019).

4. Literature review and developing hypotheses

4.1. Corporate social responsibility and operating cash flow opacity

The association between CSR and OCFOP in the literature is discussed by deploying different perspectives and theories. Many studies have stressed the importance of institutional and organizational factors as they directly affect CSR practices. These factors impact managers and their activities to enhance equality, embrace justice and product development, or they may behave in an opportunistic way and work for their own benefits (Brahem et al., 2022; Sial et al., 2018). Moreover, firm opacity theory states that most firms seek profit maximization. While seeking this goal, firms use CSR practices as signals for their ethical practices and high quality performance, and to mitigate the information asymmetry in their reporting (Agnes Cheng et al., 2020; Ball et al., 2000; Hamrouni et al., 2021; Prior et al., 2008; You & Chen, 2022). The impact of this signaling is more profound when the firm has high levels of information asymmetry or opacity (You & Chen, 2022). In that sense, opacity focuses on the possibility of misleading potential and existing investors of the company's true financial position by diminishing information movement from insiders to outsiders, delivering a company's mysterious financially healthy situation (Kim & Yoo, 2022). In the current study, we used the OCFOP as an opacity detection building on the Agnes Cheng et al. (2020) model for measuring opacity. The OCFOP is essential in assessing a firm's performance by supporting managers with masks to keep back bad news, making it easier for them to misuse the company's resources (Agnes Cheng et al., 2020).

Numerous studies within the extant literature have extensively investigated the link between CSR and earnings management. These studies have yielded a diverse array of findings, underscoring the lack of consensus within the body of research (Abdelfattah & Elfeky, 2021; Brahem et al., 2022; Grougiou et al., 2019; Lee, 2012; Liu & Lee, 2019; Mohamed et al., 2019; Mohmed et al., 2019; Nguyen, 2022; Prior et al., 2008; Sial et al., 2018). Theoretically, the association between CSR and OCFOP has been connected to agency theory by many studies in the literature. Agency-based studies report a positive relationship between CSR and OCFOP (Muttakin & Subramaniam, 2015; Prior et al., 2008; Rahman & Zheng, 2022; Salewski & Zülch, 2014; Sial et al., 2018).

Based on agency theory, those studies explained that the management (agent) has access to the information available, and they may behave to utilize the whole information for their own remunerations and other benefits instead of supporting the owners' (stockholders) benefits (Fama & Jensen, 1983; Kılıc & Kuzey, 2018). Opacity causes poor transparency in financial reporting, thereby increasing agency costs and raising managers' opportunistic behavior. As a result of this opportunistic behavior, managers may use their engagement in CSR practices to present a good image of the company's reporting and thus hide their true company's performance. At this moment, CSR would act to provide a diversion tactic rather than presenting an initiative for further voluntary disclosure and subsequently reducing agency costs (Abdelfattah & Elfeky, 2021; Jouber, 2019b; Mohmed et al., 2019; Rahman & Zheng, 2022). Likewise, Abdelfattah and Elfeky (2021) empirical findings in the Egyptian market suggest a positive relationship between CSR practices and earnings management, in addition to Mohmed et al. (2019) finding that most Egyptian listed firms used to hide their poor earnings quality by investing in CSR practices.

In contrast, studies that report a negative relationship between CSR and OCFOP have mainly deployed two theories: stakeholder theory and stewardship theory (Cho et al., 2013; Chulkov & Wang, 2023; Cui et al., 2018; Kim & Yoo, 2022; Nair et al., 2019). Stakeholder theory views the most essential task for a firm to carry out stakeholder expectations and, therefore, meet their interest toward the most efficient use of the firm's resources (Grougiou et al., 2019). Based on the theory's concepts, it is expected that companies highly engaged in CSR activities would be less corporate tax avoidant and imply less aggressive accounting, which limits the opacity of financial reporting and enhances financial transparency and reporting quality by employing fewer earnings opacities (Abdul Wahab et al., 2018; Cho et al., 2013; Cui et al., 2018; Nair et al., 2019). Firms are recommended to disclose more information regarding their CSR practices, which enhances financial transparency and reporting quality by employing fewer earnings opacities (Nair et al., 2019).

Moreover, stewardship theory assumptions support the negative relationship between CSR and OCFOP, as corporate managers are expected to perform as optimal agents for their shareholders and, therefore, would likely maximize their interest in better-making shareholders' economic decisions (Mohmed et al.,

2019). Accordingly, based on academic research empirically found that theories, a negative relationship exists between CSR and opacity (Brahem et al., 2022; Chulkov & Wang, 2023; Hamrouni et al., 2021; Kim & Yoo, 2022). The same suggestions for such inverse relationships have been empirically revealed in emerging economies such as Egypt, India and East Asian markets (Aboud & Diab, 2018; Nair et al., 2019; Nguyen, 2022). Furthermore, firm opacity theory proposes that, unlike firms in developed markets, firms in less developed markets are motivated to signal their CSR activities as a reflection of financial statement transparency, and CSR practices signaling is more profound as these markets lake the essential governance tools, awareness and include a high level of information asymmetry and/or opacity (Aboud & Diab, 2018; Metwally, 2022; Metwally et al., 2021; You & Chen, 2022).

In summary, based on theoretical perspectives and previous literature explaining the relationship between CSR practices and OCFOP, we can conclude that stakeholder theory and stewardship theory-based studies explain the negative relationship between CSR practices and OCFOP by clarifying that managers would act toward maximizing their agents' interests. Conversely, firm opacity theory and agency theory-based studies explain the positive relationship between CSR and OCFOP through management's opportunistic behavior and maximizing their own benefits by producing opacity and/or asymmetry to make their CSR practices more profound in their signaling. Given the divergent findings within the existing body of literature, we postulate the presence of a potential correlation between CSR and the degree of OCFOP in the Egyptian market, refraining from making anticipatory assertions regarding the specific nature of this correlation. Consequently, we articulate our initial hypothesis as follows.

H1. There is an association between CSR and the level of operating cash flow opacity.

4.2. Moderation effect of tax avoidance between CSR and OCFOP

Several studies concentrated on the association between CSR and tax avoidance and reported contradictory results (Abdelfattah & Aboud, 2020; Abdul Wahab et al., 2022; Abid & Dammak, 2021; Chouaibi et al., 2021; Du & Li, 2023; Gavious et al., 2022; Hoi et al., 2013; Jiang et al., 2022; Kacem & Brahim Omri, 2021; Liu & Lee, 2019; Rakia et al., 2023; Richardson et al., 2013; Van & Ly, 2021; Zeng, 2019). Engaging heavily in CSR practices reduces profits because activities such as community support initiatives are often viewed as donations (Flammer, 2015). Furthermore, CSR efforts can put companies at a disadvantage compared to non-CSR competitors, as they might involve investments in environmental protection equipment, which increases costs without an immediate return. Hence, from an agency theory perspective, managers' opportunistic behavior is used to explain how managers tend to increase CSR practices and disclosure to hide underlying tax savings through tax avoidance (Gavious et al., 2022; Hoi et al., 2013; Jiang et al., 2022; Van & Ly, 2021).

On the contrary, from the agency perspective, stakeholder theory and legitimacy theorists challenge this view and argue that firms engaged in CSR can become more profitable (Van & Ly, 2021). This is due to various external factors, such as improved legitimacy, a better reputation and increased value, as well as internal factors, such as enhanced human resources (Kacem & Brahim Omri, 2021; Liu & Lee, 2019). In line with this argument, advocates of the significance of a company's resources suggest that CSR investments can help build a positive corporate reputation and enhance investment efficiency (Branco & Rodrigues, 2006; Chouaibi et al., 2021; Du & Li, 2023; Hoi et al., 2013; Lin et al., 2021; Liu & Lee, 2019; Van & Ly, 2021).

Tax avoidance and opacity association were the concerns of other studies in the literature; these studies achieved mixed results (Arman & Mira, 2021; Balakrishnan et al., 2018; Chen & Lin, 2017; Lastiati, 2020; Lee & Bose, 2021; Sá & Alves, 2022; Wang, 2010). Some studies have found a negative association between firm opacity and tax avoidance (Arman & Mira, 2021; Balakrishnan et al., 2018; Wang, 2010). Other studies found a positive association between tax avoidance and firm opacity (Chen & Lin, 2017; Kerr, 2013; Lastiati, 2020; Lee & Bose, 2021; Sá & Alves, 2022). Theoretically, the negative and positive impacts are explained by deploying the agency theory. The supporters of the negative association concentrated on managers' opportunistic behavior and how their intentions to engage in tax avoidance push them to disclose more information (e.g. management reports, CSR and other narrative disclosures) to hide their tax aggressiveness (Arman & Mira, 2021; Balakrishnan et al., 2018; Wang, 2010). On the

contrary, studies that reported positive relationships explained that managers tend not to disclose information regarding their tax plans and produce an opaque environment to hide the information and keep it for themselves (Chen & Lin, 2017; Kerr, 2013; Lastiati, 2020; Lee & Bose, 2021; Sá & Alves, 2022). However, there is a notable scarcity in the examination of the moderating influence of corporate tax avoidance on the interplay between CSR and OCFOP. Nevertheless, antecedent research underscores an established connection between tax avoidance, CSR and OCFOP. Aligning with the theoretical discourse and precedent empirical evidence, it is anticipated that tax avoidance will exert a discernible influence on the relationship between CSR and OCFOP within the Egyptian market. Accordingly, we propose the following hypothesis:

H2. Corporate tax avoidance moderates the relationship between CSR and operating cash flow opacity.

5. Research design

5.1. Identifying sample and sources

The initial sample selected for the current research was based on the top companies listed on the Egyptian Exchange, representing EGX 100. However, according to the aim of the research, which is to investigate the association between CSR and OCFOP, the sample was limited to companies listed on the S&P/EGX ESG index for 10 consecutive years, covering the period from 2012 until the end of 2021. The index started to publish only the top 30 Egyptian firms since 2007 marking their best performance within three factors environmental, social and corporate governance responsibility in comparison to other firms (Mohmed et al., 2019). However, the researchers obtained data for the top 100 listed companies that were highly committed to CSR practices. Any companies with missing data because they did not include CSR scores and had negative cash effective tax rates, in addition to insurance and financial companies, were excluded because of their special characteristics in accounting and financial reporting (Abdelazim et al., 2023; Diab et al., 2023). The final sample consists of 52 companies, making 520 observations representing 10 industries, as described in detail in Table 1. Data in this research were collected from annual reports, company websites, Bloomberg Asharq and Mubasher Egypt.

5.2. Research methodology: variables and regression models definition

We used Pooled OLS and random effect regression analysis with robust standard errors to mitigate the concerns of heteroscedasticity and autocorrelation, to investigate the relationship proposed according to the research hypotheses. Therefore, two regression models were generated to formulate the two hypotheses as follows:

Model I:

$$OCFOP_{it} = \alpha + \beta_{it} CSR + \beta_{it} LEV + \beta_{it} SIZE + \beta_{it} ROE + \varepsilon_{it}$$

Table 1. Sample classification and industry type.

7 71		
Number of companies	Number of observations	Percentage, %
3	30	5.8
5	50	9.6
10	100	19.2
3	30	5.8
4	40	7.7
12	120	23
5	50	9.6
8	80	15.4
2	20	3.9
52	520	100
	Number of companies 3 5 10 3 4 12 5 8 2	Number of companies Number of observations 3 30 5 50 10 100 3 30 4 40 12 120 5 50 8 80 2 20

Model II:

$$OCFOP_{it} = \alpha + \beta_{it} CSR + \beta_{it} CETR + \beta_{it} CSR * CETR + \beta_{it} LEV + \beta_{it} SIZE + \beta_{it} ROE + \varepsilon_{it}$$

where OCFOP represents the operating cash flows opacity; CSR denotes the reported corporate social responsibility score as disclosed in the S&P/EGX ESG index, CETR symbolizes tax avoidance, LEV denotes leverage, SIZE denotes company size, ROE symbolizes return on equity and CSR*CETR is the interaction variable, which is the outcome of the independent and moderator variables.

5.3. Variables identification and measurements

5.3.1. Dependent variable: operating cash flows opacity

As supposed by Dechow et al. (1998) we use the following model to estimate the value of abnormal operating cash flows as a measurement for operating cash flow management, as implemented by Agnes Cheng et al. (2020) and Lee (2012) using each company-level annual data over time series for the prior ten years:

$$\frac{\mathsf{OCFOP}_{\mathsf{t}}}{\mathsf{TA}_{\mathsf{t}-1}} \!=\! \lambda_{\mathsf{0}} + \lambda_{\mathsf{1}} \big(\mathsf{1/TA}_{\mathsf{t}-1} \big) + \lambda_{\mathsf{2}} \big(\mathsf{SALE}_{\mathsf{t}} \, / \, \mathsf{TA}_{\mathsf{t}-1} \big) + \lambda_{\mathsf{3}} \big(\Delta \mathsf{SALE}_{\mathsf{t}} \, / \, \mathsf{TA}_{\mathsf{t}-1} \big) + \varepsilon_{\mathsf{t}}$$

where OCFOP is the operating cash flow for period t, TA is the total assets for the ending period of t-1 and SALE and Δ SALE represent the reported total sales and the change in reported total sales during period t. Therefore, we used the parameter estimate to expect a deviation from normal OCFOP, so the residuals from the estimation model denote the difference between the normal expected levels of OCFOP. Therefore, higher values of OCFOP imply that companies are more likely to manage reported OCF, thus, having higher levels of operating cash flows opacity.

5.3.2. Independent variable: corporate social responsibility

Earlier studies employ several approaches to gauge CSR, typically combining both corporate governance (CG) and CSR into CSR measurement. However, Kim et al. (2012) separate the impact of CSR and CG by removing the CG score to calculate their CSR scores. Having a closed pointer toward the Egyptian market, Mohmed et al. (2019) have researched into the environmental and social dimensions of CSR within the Egyptian S&P/ESG index. These CSR dimensions, namely community, environment, employees and consumers/products, are derived from data within the S&P/ESG index, which encompasses environmental, social and governance scores. Other studies such as Bozzolan et al. (2015), Muttakin and Subramaniam (2015) and Salewski & Zülch (2014) base their CSR measurements on three common dimensions: society, employees and the environment. Therefore, in proportion to preceding readings and for the current research purpose, we used the composite score to measure CSR as computed according to the S&P/EGX ESG index, as suggested by Kim et al. (2012) and Mohmed et al. (2019).

5.3.3. Moderator variable: tax avoidance

Following the literature, we use the cash effective tax rate to measure tax avoidance (Jiang et al., 2022). This measurement is used to explain that the cash amount of tax would help reduce the likely effect of assessment allowance according to Dyreng et al. (2008) thus, we computed using based on the following equation:

$$CETR_t = \frac{Cash Tax Paid_t}{Pre Income Tax.}$$

where CETR is the cash effective tax rate for period t, cash tax paid is the amount of taxes paid as disclosed in the statement of cash flows for period t, and pre-income tax is earnings before taxes, as computed in the income statement for period t.

5.3.4. Control variables

In this research, we controlled for companies' leverage, firm size and return on equity, which are expected to influence the firm's level of OCFOP. Leverage is the ratio of a firm's total debt to its total assets, while firm size is measured by the log of a firm's total assets. We measure the return on equity by dividing each company's total equity by its net income (Bozzolan et al., 2015; Eliwa et al., 2016; Francis et al., 2005; Muttakin & Subramaniam, 2015). The following Table 2 summarizes all variables used in the current study.

6. Empirical results

6.1. Descriptive statistics

Table 3 presents a descriptive analysis of all the variables tested in the study. As can be seen from the table, CSR recorded an average score of 4.8157, associated with a median of 4.8153 and standard deviation of 0.07131. The minimum CSR score in our sample was 4.69 and the maximum was 4.99. CETR reported an average score of 14.23% and ranges between 0 and 1 following prior studies we deleted observations with negative CETR and further CETR of companies result in greater than one was normalized to one (Abdelfattah & Aboud, 2020; Dyreng et al., 2010; Hope et al., 2013), besides CETR was associated with a median of 9.6% and a standard deviation of 0.2184. The OCFOP reported a mean (median) of 4.62% (2.85%) and varied between 0.082% and 27.43% with a standard deviation of 5.14%.

Furthermore, controls indicated average leverage, size and ROE of 42%, 21.5% and 16.6%, respectively, with medians of 38.7%, 21.6% and 11.6%, respectively. Leverage indicated a standard deviation of 0.23, while size was 1.72, and ROE represented a standard deviation of 0.27. In addition, we noticed a significant difference in the companies' size included in the sample, as it reported a range between a minimum of 17 and a maximum of 24.77.

6.2. Correlation analysis

Based on the analysis shown in Table 4, the correlation matrix for the current research variables indicates some significant and non-significant relationships. As presented in the table a positive association occurs between the company's OCFOP and CSR at p < 0.01 of significance level and also a positive association between the company's OCFOP and CETR at p < 0.01 significance level besides a significant positive

Table 2. Summary of variables identification.

Variables	Code	Definition	Measurement
Operating cash flow opacity	OCFOP	Because of accruals in the form of estimates and judgments, management tried to manage operating cash flows	Measuring abnormal operating cash flows estimated using the residuals of the Dechow et al. (1998) model
Corporate social responsibility	CSR	CSR score was according to the index following each firm's practices of community, environment, employees and consumer and product	The natural logarithm of social and environmental reporting is the total of four categories measured by the Egyptian ESG index (community, environment, employees and consumer and product) for the firm i in the year t
Tax avoidance	CETR	The act to reduce taxes paid to the minimum level	Cash tax paid divided by pre-income tax
Leverage	LEV	Use of borrowings to fund assets acquisition	Total debt is divided by total assets.
Size	SIZE	Natural logarithm of total assets.	Log of assets
Return on equity	ROE	Reference for the return on the total amount of equity	Net income divided by total equity

Table 3. Descriptive statistics.

Variables	Mean	Median	SD	Min	Max
OCFOP	0.0461852	0.0285461	0.0513805	0.0008292	0.2743154
CSR	4.815743	4.815324	0.0713068	4.68907	4.9985
CETR	0.1422681	0.0957809	0.2183579	0.00000	0.833961
LEV	0.4182298	0.3866356	0.233507	0.0251077	1.052822
SIZE	21.47447	21.62185	1.721356	17.04227	24.77042
ROE	0.1660711	0.1155342	0.2687309	-0.3442261	1.664038

association between OCFOP and LEV at p < 0.05. However, a significant negative association was observed between OCFOP and size (p < 0.01). A positive relationship emerges between CSR and both SIZE and ROE at p < 0.01 level of significance while no significant association exists between CSR, CETR and LEV. Furthermore, a positive relationship arises between CETR and both LEV and SIZE at a significance level of p < 0.10 and p < 0.01, respectively. In addition, a significant positive relationship occurs between LEV and SIZE at p < 0.01, while a significant negative association exists between LEV and ROE at p < 0.01. Finally, a significant positive correlation was observed between the SIZE and ROE (p < 0.05). Furthermore, the analysis showed that there was no multicollinearity between the independent variables; coefficients were not higher than 0.34 (). Also, to make a double checking for the absence of multicollinearity, the variance inflation factor (VIF) reported the highest observed value of 1.32 which is below the conventional cut off of 10.0 besides Mean VIF stated value of 1.14 which is less than 2 (Rakia et al., 2023).

6.3. Findings and discussions

As demonstrated in Tables 5 and 6, the analysis revealed from the pooled ordinary least squares OLS regression and the random effects of the regression model are summarized for firms' operating cash flow opacity, including corporate social responsibility, leverage, size and return on equity, as indicated in Model I. The next step shows the pooled OLS regression and the random effects regression model of firms' OCFOP involving corporate social responsibility, cash effective tax rate, leverage, size and return on equity, as implied based on Model II.

Therefore, based on the pooled OLS and random effects results indicated in Model I of Table 5. The pooled OLS estimator does not consider the panel structure of the data, leading to a potential shortcoming in the procedure of biased estimates (Cameron & Trivedi, 2005). Accordingly, the Hausman test indicated that random effects are preferred over the fixed effects model estimator ($x^2 = 0.56$, Prob > x^2 = 1.000); hence, a random effects model is also projected (Boshnak, 2021). The table demonstrates that a positive relationship exists between the independent variable CSR and the dependent variable OCFOP, as the coefficient was 0.129 for the pooled model and 0.068 for the random effects model, which is significant for both models, as indicated by p < 1% and p < 10%, respectively. Likewise, based on the pooled OLS model, controls, such as LEV and ROE, show a positive association alongside the dependent variable OCFOP with coefficients of 0.045 and 0.018, and both were significant at p < 1% and p < 5%, respectively; however, from the random effects model, only LEV demonstrates a positive association with the dependent variable OCFOP with a coefficient of 0.049, which is significant at the 1% level. However, SIZE denotes a significant negative association per OCFOP, with a coefficient of-0.008 for the pooled OLS

Table 4. Correlation matrix.

Variables	OCFOP	CSR	CETR	LEV	SIZE	ROE	VIF
OCFOP	1.0000						NA
CSR	0.1222***	1.0000					1.12
CETR	0.1778***	0.0458	1.0000				1.05
LEV	0.1127**	0.0619	0.0802*	1.0000			1.17
SIZE	-0.1333***	0.3106***	0.2120***	0.3399***	1.0000		1.32
ROE	0.0655	0.1384***	-0.0379	-0.1318***	0.1032**	1.0000	1.06
Mean VIF 1 14							

For significant correlation indication, *** p < 0.01, ** p < 0.05 and * p < 0.10.

Table 5. Results of regression analysis – Model I.

Variables	Pooled	d OLS regression r	nodel	Random effects regression model		
	Coef	Z	P>Z	Coef	Z	P>Z
CSR	0.1295585	4.03	0.000***	0.0681275	1.74	0.082*
LEV	0.0452244	4.52	0.000***	0.0498479	4.23	0.000***
SIZE	-0.0080252	-5.67	0.000***	-0.074818	-3.71	0.000***
ROE	0.0182556	2.21	0.028**	0.00683	0.70	0.484
Cons	-0.4273445	-2.88	0.004***	-0.1432141	-0.79	0.431
Adjusted R ²	$rd R^2$ 0.0801 0.2043					
<i>F</i> -value		12.31***			25.06***	
Observation		520			520	

Notes: ***, ** and * reflect significance at 1, 5 and 10% levels, respectively.



Table 6. Results of regression analysis – Model II.

	Pooled	d OLS regression r	nodel	Random effects regression model		
Variables	Coef.	Z	P>Z	Coef.	Z	P>Z
CSR	0.1275501	4.06	0.000***	0.0697886	1.84	0.066*
CETR	0.0571337	5.58	0.000***	0.0534501	5.21	0.000***
CSR*CETR	0.2328622	1.99	0.052*	0.2691652	1.86	0.063*
LEV	0.0445705	4.57	0.000***	0.0479674	4.18	0.000***
SIZE	-0.009396	-6.68	0.000***	-0.0087985	-4.50	0.000***
ROE	0.0216879	2.68	0.008***	0.0105676	1.12	0.265
Cons	-0.396823	-2.74	0.006***	-0.1305659	-0.74	0.458
Adjusted R ²		0.1395			0.2825	
<i>F</i> -value		13.86***			54.70***.	
Observation		520			520	

Notes: ***, ** and * reflect significance at 1, 5 and 10% levels, respectively.

model and -0.075 for the random effects model, and this association is significant for both models at the p < 1% level.

Moreover, the two models illustrate F-value equal to 12.31 and 25.06 respectively and both models were significant at the 1%; this significant positive association indicates that the whole model is significant. In addition, the pooled model had an adjusted R^2 of 0.0801 and a significant F-value (F=12.31, p < 0.001), indicating that the independent variable CSR and the controls LEV, SIZE and ROE could explain an 8.01 percentage of variation in the dependent variable OCFOP. Moreover, the random-effects model had an adjusted R^2 of 0.2043 and a significant F-value (F=25.06, p<0.001). The findings of the previous analysis support the acceptance of the first hypothesis H₁ and reject the null hypothesis H₀ making it clear that companies with higher CSR deliver a higher rate of cash flow opacity.

The above results support agency theory assertions regarding managers' opportunistic behavior resulting from information asymmetry. Therefore, managers are incentivized to act in their own interests rather than in the interests of their companies' shareholders (Mohmed et al., 2019). These results are consistent with those of previous studies (Muttakin & Subramaniam, 2015; Prior et al., 2008; Rahman & Zheng, 2022; Salewski & Zülch, 2014; Sial et al., 2018). Those who concluded a similar positive relationship between CSR and OCFOP indicated a reference to opportunistic behavior by the company's management, intending to conceal their manipulation of financial reporting and not transparently disclose the true financial position of the company. In contrast, it disagrees with other studies that report either a negative or no significant relationship between CSR and OCFOP (Aboud & Diab, 2018; Brahem et al., 2022; Chulkov & Wang, 2023; Hamrouni et al., 2021; Kim & Yoo, 2022; Nair et al., 2019; Nguyen, 2022). Thus, the results of the current study do not support stakeholder and stewardship theories. Managers of Egyptian non-financial firms did not perform as optimal agents for their shareholders, and they tended to be less transparent in their disclosures.

In Table 6, the pooled OLS and random effects results indicated in Model II show a positive relationship between the independent variable CSR and the dependent variable OCFOP, as the coefficient was 0.128 for the pooled OLS model and 0.069 for the random effects model, which are significant at the 1% and 10% levels, respectively. Similarly, the findings revealed a positive relationship between the moderator CETR and the dependent variable OCFOP, with a coefficient of 0.0571 for the pooled OLS model and 0.0535 for the random effects model, which is significant at p < 1% for both models, thus indicating the effect of CETR as an independent variable. Additionally, based on the pooled OLS model, the interaction of CSR*CETR shows a positive relationship with the dependent variable OCFOP, with a coefficient of 0.052 and a coefficient of 0.063 based on the random effects model, which is significant at p < 10% for both models. Conversely, based on the pooled OLS model, a negative relationship exists between SIZE as a control and the dependent variable OCFOP with a coefficient of -0.009, and from the random effects model with a coefficient of -0.008, both of which were significant at the 1% level. However, other controls, such as LEV and ROE, deliver a positive association with the dependent variable OCFOP with a coefficient for LEV of 0.0446 for the pooled OLS model and 0.048 for the random effects model, both of which were significant at the 1% level. In addition, the ROE coefficient of 0.0217 was significant only for the pooled OLS model at the 1% level.

Furthermore, the two models illustrate F-value equal to 13.86 and 54.70 respectively and both models were significant at the 1% level; this significant positive association indicates that the whole model is significant. In addition, the findings in Table 6 indicate that the pooled model has an adjusted R^2 of 0.14 and a significant F-value (F=13.86, p<0.001), supporting that the independent variable CSR, moderator CETR and the controls LEV, SIZE and ROE can interpret 14 percent of the variation in the dependent variable OCFOP. In addition, the random-effects model had an adjusted R² of 0.283 and a significant F-value (F=54.70, p<0.001). As a result of the previous analysis, the findings support the acceptance of the second hypothesis H₂ and reject the null hypothesis H₀, assuming that companies with higher CSR moderated with higher values of CETR generate a higher level of cash flow opacity. This implies that corporate tax avoidance has a positive moderating role as it strengthens the effect of CSR on OCFOP.

The results presented above substantiate the assertions of agency theory concerning managers' OB. This consequently portrays CSR practices as a form of greenwashing for the company's aggressive tax strategies, which mirrors their lack of transparency in financial disclosures. These findings align with earlier research that interpreted a direct connection between CSR and tax avoidance, revealing a consistent positive correlation (Abdelfattah & Aboud, 2020; Abid & Dammak, 2021; Gavious et al., 2022; Jiang et al., 2022; Marwat et al., 2021; Zeng, 2019). As these companies achieve tax savings while providing increased CSR information, CSR practices incentivize managers to prioritize tax savings, leading to reduced transparency and heightened opacity in their financial reporting (Lee & Bose, 2021). On the contrary, the current study's results contradict the results of some studies, as these studies concluded a negative association between firms' transparency and tax avoidance (Arman & Mira, 2021; Lastiati, 2020; Sá & Alves, 2022; Wang, 2010). That said, the results of this study do not align with the assertions of stakeholder and legitimacy theories, which suggest that firms engaging in CSR are persuaded to uphold social prosperity. Consequently, tax avoidance may be seen as a manifestation of management's neglectful behavior toward the company's reputation and stakeholders' interests of its stakeholders (Chouaibi et al., 2021; Hoi et al., 2013; Van & Ly, 2021).

7. Additional analyses

As an additional test, we investigated whether the relationship between corporate social responsibility and operating cash flow opacity is subject to change because of other factors by employing more control variables including, tangibility which is measured by dividing property, plant and equipment to total assets based on Dietrich (2007), return on assets which is an accounting-based indicator measured by dividing net income over firm's total assets as stated by Diab et al. (2024), cash proportion of cash and short-term investment to total assets as cited by Cao et al. (2022) and selling, general and administrative expenses as proportion to net sales as mentioned by Beneish (1999). As a result, to additionally examine the association of CSR on firm opacity, the previous models are reformulated as follows:

Model I:

OCFOP_{ir} =
$$\alpha + \beta_{ir}$$
CSR + β_{ir} LEV + β_{ir} SIZE + β_{ir} ROE + β_{ir} TANG + β_{ir} ROA + β_{ir} CASH + β_{ir} SGA + ε_{ir}

Model II:

$$\begin{aligned} \mathsf{OCFOP_{it}} &= \alpha + \beta_{it} \, \mathsf{CSR} + \beta_{it} \mathsf{CSR} * \mathsf{CETR} + \beta_{it} \mathsf{LEV} + \beta_{it} \mathsf{SIZE} \\ &+ \beta_{it} \, \mathsf{ROE} + \beta_{it} \, \mathsf{TANG} + \beta_{it} \, \mathsf{ROA} + \beta_{it} \, \mathsf{CASH} + \beta_{it} \, \mathsf{SGA} + \varepsilon_{it} \end{aligned}$$

where:

 $OCFOP_{it}$ is the operating cashflows of firm *i* for period *t*. CSR is the corporate social responsibility score as disclosed in the S&P/EGX ESG index LEV is the leverage SIZE is the company size measured by log of total assets ROE is the return on equity TANG is the tangibility ROA is the return on assets



Table 7. Results of regression analysis using additional control variables – Model I.

	Poole	d OLS regression m	odel	Random effects regression model		
Variables	Coef	Z	P>Z	Coef	Z	P>Z
CSR	0.1378	4.27	0.000***	0.0748	1.93	0.054*
LEV	0.0347	2.99	0.003***	0.0337	2.56	0.010**
SIZE	-0.0078	-4.54	0.000***	-0.0065	-2.88	0.004***
ROE	0.02464	2.48	0.013**	0.0171	1.57	0.116
ROA	-0.0291	-2.29	0.022**	-0.326	-2.18	0.029**
TANG	-0.0268	-3.18	0.002***	-0.0221	-2.06	0.040**
Cash	-0.0323	-1.74	0.083*	-0.0424	-1.98	0.048**
SAG	-0.4507	-1.38	0.167	-0.0043	-0.27	0.784
Cons	-0.4507	-3.05	0.002***	-0.1757	-0.98	0.328
Adjusted R ²		0.105			0.252	
<i>F</i> -value		8.65***			39.81***	
Observation		520			520	

Notes: ***, ** and * reflect significance at 1, 5 and 10% levels, respectively.

Table 8. Results of regression analysis using additional control variables – Model II.

	Poole	ed OLS regression	model	Rando	m effects regressior	n model
Variables	Coef.	Z	P>Z	Coef.	Z	P > Z
CSR	0.1332	4.23	0.000***	0.0749	2.00	0.046**
CETR	0.0584	5.74	0.000***	0.05441	5.34	0.000***
CSR*CETR	0.2172	1.89	0.062*	0.2379	1.75	0.082*
LEV	0.0307	2.72	0.007***	0.0308	2.40	0.016**
SIZE	-0.0085	-5.07	0.000***	-0.0074	-3.40	0.001***
ROE	0.0309	3.19	0.001***	0.0223	2.10	0.036**
ROA	-0.0351	-2.84	0.005***	-0.0375	-2.58	0.010**
TANG	-0.248	-3.03	0.003***	-0.0208	-2.01	0.044**
Cash	-0.3001	-1.66	0.098*	-0.0362	-1.74	0.085*
SAG	-0.0112	-0.79	0.430	-0.0011	-0.17	0.745
Cons	0.2273	6.35	0.000***	0.2028	4.35	0.000***
Adjusted R ²		0.156			0.343	
F-value		10.63***			71.43***.	
Observation		520			520	

Notes: ***, ** and * reflect significance at 1, 5 and 10% levels, respectively.

Cash is the cash holding

SGA is the selling, general and administrative expenses

CSR*CETR is the interaction variable which is the outcome of the independent and the moderator.

Table 7 presented a positive relationship exists between the independent variable CSR and the dependent variable OCFOP, as the coefficient was 0.1378 for the pooled model and 0.0748 for the random effects model, which is significant for both models, as indicated by p < 1% and p < 10%, respectively. Furthermore, based on the pooled OLS model, additional controls interacted, such as ROA, TANG and CASH, show a negative relationship alongside the dependent variable OCFOP with coefficients of -0.0291, -0.0268 and -0.0323, and were significant at p < 5%, p < 1% and p < 10%, respectively; however, from the random effects model, TANG and CASH demonstrates a significant at the 5% level. Likewise, the two models illustrate F-value equal to 8.65 and 39.81, respectively, and both models were significant at the 1%; this significant positive association indicates that the whole model is significant.

Table 8 shows the pooled OLS and random effects results indicated in the reformed Model II illustrates a positive relationship between the independent variable CSR and the dependent variable OCFOP, as the coefficient was 0.1332 for the pooled OLS model and 0.0749 for the random effects model, which are significant at the 1% and 5% levels, respectively. Similarly, the findings revealed a positive relationship between the moderator CETR and the dependent variable OCFOP, with a coefficient of 0.0584 for the pooled OLS model and 0.05441 for the random effects model, which is significant at p < 1% for both models, thus, indicating the effect of CETR as an independent variable. Additionally, based on the pooled OLS model, the interaction of CSR*CETR shows a positive relationship with the dependent variable OCFOP, with a coefficient of 0.2172 and a coefficient of 0.2379 based on the random effects model, which is significant at p < 10% for both models.

Conversely, adding the new controls, as ROA, TANG and CASH, the pooled OLS model shows a negative relationship alongside the dependent variable OCFOP with coefficients of -0.0351, -0.248 and -0.3001, and were significant at p < 1%, p < 1% and p < 10%, respectively; furthermore, from the random effects model, TANG and CASH demonstrates a significant at the 5% level. Similarly, the two models illustrate F-value equal to 10.63 and 71.43, respectively, and both models were significant at the 1%; this significant positive association indicates that the whole model is significant.

As a result, the additional analysis provided by employing extra control variables using the reformed model 1 and 2 yields outcomes consistent with the previous findings suggesting a positive significant relationship between corporate social responsibilities and operating cash flow opacity. Additionally, there is a suggestion of the significance of the moderating effect of the corporate tax avoidance variable.

8. Robustness analysis

8.1. Exploiting another measurement of the dependent variable

To validate the findings of the current study, an alternate measurement of the dependent variable was used. Thus, another measurement was used to measure OCFOP, we use the absolute value of the change in cash conversion cycle measured as the absolute value of the change in days in accounts receivable plus days in inventory minus days in accounts payable from the fourth quarter in year t-1 (ΔCC) as suggested by Agnes Cheng et al. (2020).

Tables 9 and 10 shows the findings of the sensitivity analysis used to evaluate the sensitivity of the results to changes in firm opacity by presenting a different dependent variable measurement $|\Delta CC|$. The findings are robust for the purpose of delivering another measurement of firm opacity, as the outcomes of both models were coherent with the prior outcomes.

8.2. The endogeneity issue

Although the current research findings indicates that there's a significant positive relationship between CSR and OCFOP besides exhibiting the moderating effect of tax avoidance, endogeneity bias may occur. Endogeneity problem is a possible matter that may appear because of omitted variables affecting the

Table 9. Results of regression analysis using another measurement for dependent variable – Model I.

	Pool	Pooled OLS regression model			Random effects regression model		
Variables	Coef	Z	P>Z	Coef	Z	P>Z	
CSR	0.9816	2.70	0.007***	1.4283	3.41	0.000***	
LEV	0.1478	1.13	0.259	0.2243	1.45	0.148	
SIZE	-0.0190	-1.06	0.290	-0.0073	-0.23	0.818	
ROE	0.1780	1.66	0.098*	0.2470	1.92	0.055*	
Cons	3.820	10.18	0.000***	4.3666	6.60	0.000***	
Adjusted R ²	0.032			0.023			
<i>F</i> -value		4.11***			17.51***		
Observation		520			520		

Notes: ***, ** and * reflect significance at 1, 5 and 10% levels, respectively.

Table 10. Results of regression analysis using another measurement for dependent variable - Model II.

	Poole	Pooled OLS regression model			Random effects regression model		
Variables	Coef.	Z	P>Z	Coef.	Z	P>Z	
CSR	0.9378	2.56	0.011**	1.4054	3.38	0.001***	
CSR*CETR	0.2368	1.81	0.071*	0.2008	1.74	0.076*	
LEV	3.9234	2.50	0.013**	5.0818	3.44	0.001***	
SIZE	0.1139	0.87	0.383	0.1759	1.14	0.255	
ROE	-0.0183	-1.00	0.318	-0.0090	-0.28	0.776	
Cons	3.843	10.18	0.000***	4.4142	6.59	0.000***	
Adjusted R ²		0.048			0.021		
F-value		4.27***			31.88***.		
Observation		520			520		

Notes: ***, ** and * reflect significance at 1, 5 and 10% levels, respectively.



Table 11. Results of regression analysis using two step GMM method.

	Mod	lel 1	Model 2		
Predictor	Coef	Z	Coef	Z	
OCFOP _{it-1}	0.3578	4.41***	0.2225	2.67***	
CSR	0.2263	2.41**	0.1302	2.45**	
CETR	_	_	0.0730	1.82*	
CSR*CETR	_	_	1.047	2.09**	
LEV	0.0163	1.95**	0.0172	1.74*	
SIZE	-0.0015	-1.66*	-0.0002	-1.13	
ROE	0.0296	2.01**	0.0067	1.96**	
Cons	0.1880	0.64	0.0257	0.84	
-value	693.9	93***	498	3.36***	
Obs.	520		520)	
AR1(p-value)	0.0	000	0.002		
AR2(p-value)	3.0	306	0.524		
Sargan test (<i>p</i> -value)	0.5	0.547 0.425			
Hansen test (p-value)	0.7	156	(0.226	

Notes: ***, ** and * reflect significance at 1, 5 and 10% levels, respectively.

relationship between our dependent and independent variable and needs to be examined using supplementary testing. Consequently, we managed to re-estimate both models 1 and 2 outlining the two-step system Generalized Method of Moments (GMM). As illustrated through Table 11, we used standard diagnostic tests like Arellano-Bond (AR) test for first-order and Second-order correlation, Sargan test and Hansen test (Ali et al., 2024; Diab et al., 2024).

As indicated in Table 11, the Arellano-Bond autocorrelation tests shows no existence of serial correlation in error terms since the p-value of the AR2 is < 0.05 in both models, besides the p-value of the Sargan and Hansen tests is < 0.05 for models 1 and 2 which propose that the instruments employed in the GMM predictor are exogenous. Thus, results analysed from Table 9 are supported with the main results obtained from the current study (i.e. remain the same with no variations due to sensitivity).

9. Conclusion, limitations and future research

A sample of 52 Egyptian non-financial institutions is listed on the EGX with 520 observations covering the period from 2012 to 2021. This study investigates the relationship between companies with CSR and OCFOP. Moreover, this study examines the moderating role of tax avoidance in the association between a company's reported CSR score and OCFOP level. Our results show that the relationship between CSR and OCFOP is positively significant, indicating that firms that practice CSR are involved in a higher level of OCFOP. This result supports the early findings of some early studies (Muttakin & Subramaniam, 2015; Prior et al., 2008; Rahman & Zheng, 2022; Salewski & Zülch, 2014; Sial et al., 2018) in their findings that CSR practices are an opportunity to hide the true financial reporting of the company toward their stockholders. To the best of our knowledge, this study is the first to investigate the moderating effects of tax avoidance on the relationship between CSR practices and OCFOP in an emerging market in general and Egypt specifically. Moreover, our findings reveal that corporate tax avoidance moderates the relationship between CSR and OCFOP as it strengthens their relationship and shows that firms engaged in CSR practices if incentivized with tax avoidance opportunities would report higher opacity in their cash flow reporting.

Although the present investigation offers a substantive contribution, it is pertinent to acknowledge certain constraints. Primarily, this study confines its scope to non-financial entities listed on the EGX in Egypt from 2012 to 2021. Nevertheless, prospective research endeavors hold the potential to replicate this study in diverse African contexts, with a particular emphasis on the MENA region, which exhibits a paucity of studies focusing on OCFOP. Such replications assume significance within the MENA region, given that many nations therein have recently experienced parallel political and economic upheaval during the Arab Spring revolution. Second, the cash effective tax rate is widely used as a proxy to measure tax avoidance in the literature; however, it has also been criticized for its accuracy in clarifying tax avoidance exists (Dyreng et al., 2008; Hope et al., 2013). The current research relies on data revealed from non-financial listed companies, which cannot be generalized because CSR reporting in Egypt includes financial and non-listed Egyptian companies, which may not represent the population of listed firms in Egypt.

This study has several implications for Egyptian companies, investors and regulators. Based on the findings of this study, it is recommended that the stockholder awareness of practices of companies' management may be involved in attaining their objectives. In addition, the current study recommends having efficient internal controls that would be reflected in limiting cash flow opacity and thus providing more transparent and valid financial reports. Moreover, the results revealed that companies must engage in future CSR practices that could commit to sustainability concepts rather than being a way diverting from weak transparency in financial reporting. In addition, both local and international investors are advised to consider the aforementioned findings when deliberating their investment choices within the realm of Egyptian listed enterprises. For example, it is suggested that any investor in the Egyptian market should augment their information search beyond the confines of annual reports and CSR disclosures, particularly when considering investments in companies exhibiting elevated levels of CSR disclosure, or those identified as deeply committed to CSR activities. Finally, the current findings support the enforcement of further investment acts that would enrich the protection of stockholders' rights, consequently making the Egyptian market more attractive for foreign investments.

In conclusion, the present study lays the foundation for future research endeavors, advocating for an expansion of the sample to encompass financial entities listed in the EGX and engaging in CSR activities correlated with levels of OCFOP. Additionally, future investigations could entail further comparative analyses across different sectors within the EGX as well as the incorporation of non-Egyptian listed firms to bolster our empirical substantiation. Furthermore, the introduction of an additional moderator, such as political affiliation, may provide additional insights into both CSR initiatives and taxation practices. Notably, the existing study omits any discourse on governance mechanisms that could potentially mitigate managers' opportunistic behaviors. In light of this, future research may focus on delineating the moderating influence of corporate governance mechanisms (e.g. board composition, board size, board meeting frequency and board gender diversity) on the interplay between CSR and OCFOP.

Authors' contributions

Conceptualization, A.B.M.M., A.A.M.E. and M.I.S.; methodology, A.B.M.M., A.A.M.E. and M.I.S.; software, A.B.M.M., A.A.M.E. and M.I.S.; validation A.B.M.M., A.A.M.E. and M.I.S.; analysis and interpretation of the data A.B.M.M., A.A.M.E. and M.I.S.; drafting of the paper A.B.M.M., A.A.M.E. and M.I.S.; critical revision for intellectual content A.B.M.M., A.A.M.E. and M.I.S.; funding acquisition, A.B.M.M. All the authors have read and agreed to the published version of the manuscript.

Disclosure statement

No potential conflict of interest was reported by the authors.

Funding

This research was funded by the Deanship of Scientific Research, Vice Presidency for Graduate Studies and Scientific Research, King Faisal University, Saudi Arabia [Project No. GRANT 5,908].

ORCID

Abdelmoneim Bahyeldin Mohamed Metwally http://orcid.org/0000-0003-1660-766X Ahmed Abdelaty M. Elsharkawy http://orcid.org/0009-0008-8682-9640 Mohamed Ibrahim Salem http://orcid.org/0009-0009-0042-6996

Data availability statement

Data is available upon request from researchers who met the eligibility criteria. The corresponding author was contacted privately through e-mail.



References

- Abdel-Mowla, S. A. A. (2012). The Egyptian tax system reforms, investment and tax evasion (2004-2008). Journal of Economic and Administrative Sciences, 28(1), 53-78, https://doi.org/10.1108/10264111211218522
- Abdelazim, S. I., Metwally, A. B. M., & Aly, S. A. S. (2023). Firm characteristics and forward-looking disclosure: The moderating role of gender diversity. Journal of Accounting in Emerging Economies, 13(5), 947-973. https://doi. org/10.1108/JAEE-04-2022-0115
- Abdelfattah, T., & Aboud, A. (2020). Tax avoidance, corporate governance, and corporate social responsibility: The case of the Egyptian capital market. Journal of International Accounting, Auditing and Taxation, 38, 100304. https:// doi.org/10.1016/j.intaccaudtax.2020.100304
- Abdelfattah, T., & Elfeky, M. (2021). Earnings management, corporate social responsibility and governance structure: Further evidence from Egypt. International Journal of Accounting, Auditing and Performance Evaluation, 17(1/2), 173-201. https://doi.org/10.1504/IJAAPE.2021.117576
- Abdelmotaleb, M., Mohamed Metwally, A. B. E., & Saha, S. K. (2018). Exploring the impact of being perceived as a socially responsible organization on employee creativity. Management Decision, 56(11), 2325-2340. https://doi. org/10.1108/MD-06-2017-0552
- Abdul Wahab, N. S., Ntim, C. G., Mohd Adnan, M. M., & Tye, W. L. (2018). Top management team heterogeneity, governance changes and book-tax differences. Journal of International Accounting, Auditing and Taxation, 32, 30-46. https://doi.org/10.1016/j.intaccaudtax.2018.07.002
- Abdul Wahab, N. S., Ntim, C. G., Tye, W. L., & Shakil, M. H. (2022). Book-tax differences and risk: Does shareholder activism matter? Journal of International Accounting, Auditing and Taxation, 48, 100484. https://doi.org/10.1016/j. intaccaudtax.2022.100484
- Abid, S., & Dammak, S. (2021). Corporate social responsibility and tax avoidance: The case of French companies. Journal of Financial Reporting and Accounting, 20(3/4), 618-638. https://doi.org/10.1108/JFRA-04-2020-0119
- Aboud, A., & Diab, A. (2018). The impact of social, environmental and corporate governance disclosures on firm value. Journal of Accounting in Emerging Economies, 8(4), 442-458. https://doi.org/10.1108/JAEE-08-2017-0079
- Agnes Cheng, C. S., Li, S., & Zhang, E. X. (2020). Operating cash flow opacity and stock price crash risk. Journal of Accounting and Public Policy, 39(3), 106717. https://doi.org/10.1016/j.jaccpubpol.2020.106717
- Alatawi, I. A., Ntim, C. G., Zras, A., & Elmagrhi, M. H. (2023). CSR, financial and non-financial performance in the tourism sector: A systematic literature review and future research agenda. International Review of Financial Analysis, 89, 102734. https://doi.org/10.1016/j.irfa.2023.102734
- Ali, M. A. S., Aly, S. A. S., Abdelazim, S. I., & Metwally, A. B. M. (2024). Cash holdings, board governance characteristics, and Egyptian firms' performance. Cogent Business & Management, 11(1), 2302205. https://doi.org/10.1080/2331197
- Alnaim, M., & Metwally, A. B. M. (2024). Institutional pressures and environmental management accounting adoption: Do environmental strategy matter? Sustainability, 16(7), 3020. https://doi.org/10.3390/su16073020
- Arman, A., & Mira, M. (2021). Does tax avoidance make do earning opacity? Atestasi: Jurnal Ilmiah Akuntansi, 4(1), 88-95. https://doi.org/10.33096/atestasi.v4i1.655
- Balakrishnan, K., Blouin, J. L., & Guay, W. R. (2018). Tax aggressiveness and corporate transparency. The Accounting Review, 94(1), 45-69. https://doi.org/10.2308/accr-52130
- Ball, R., Kothari, S. P., & Robin, A. (2000). The effect of international institutional factors on properties of accounting earnings. Journal of Accounting and Economics, 29(1), 1-51. https://doi.org/10.1016/S0165-4101(00)00012-4
- Beneish, M. D. (1999). The detection of earnings manipulation. Financial Analysts Journal, 55(5), 24-36. https://doi. org/10.2469/faj.v55.n5.2296
- Boshnak, H. A. (2021). Corporate governance mechanisms and firm performance in Saudi Arabia. International Journal of Financial Research, 12(3), 446-465. https://doi.org/10.5430/ijfr.v12n3p446
- Bozzolan, S., Fabrizi, M., Mallin, C. A., & Michelon, G. (2015). Corporate social responsibility and earnings quality: International evidence. The International Journal of Accounting, 50(4), 361-396. https://doi.org/10.1016/j.intacc.2015.10.003
- Brahem, E., Depoers, F., & Lakhal, F. (2022). Corporate social responsibility and earnings quality in family firms. Journal of Applied Accounting Research, 23(5), 1114-1134. https://doi.org/10.1108/JAAR-05-2021-0139
- Branco, M. C., & Rodrigues, L. L. (2006). Corporate social responsibility and resource-based perspectives. Journal of Business Ethics, 69(2), 111-132. https://doi.org/10.1007/s10551-006-9071-z
- Bremer, J., & Elias, N. (2007). Corporate governance in developing economies? The case of Egypt. International Journal of Business Governance and Ethics, 3(4), 430-445. https://doi.org/10.1504/IJBGE.2007.015210
- Buhaya, M. I., & Metwally, A. B. M. (2024). Green intellectual capital and green supply chain performance: Do external pressures matter? Cogent Business & Management, 11(1), 2349276. https://doi.org/10.1080/23311975.2024.2349276
- Cameron, A. C., & Trivedi, P. (2005). Microeconometrics: Methods and applications. Cambridge University Press.
- Cao, Z., Chen, S. X., Harakeh, M., & Lee, E. (2022). Do non-financial factors influence corporate dividend policies? Evidence from business strategy. International Review of Financial Analysis, 82, 102211. https://doi.org/10.1016/j. irfa.2022.102211
- Chen, T., & Lin, C. (2017). Does information asymmetry affect corporate tax aggressiveness? Journal of Financial and Quantitative Analysis, 52(5), 2053-2081. https://doi.org/10.1017/S0022109017000576

- Cho, S. Y., Lee, C., & Pfeiffer, R. J. (2013). Corporate social responsibility performance and information asymmetry. Journal of Accounting and Public Policy, 32(1), 71-83. https://doi.org/10.1016/j.jaccpubpol.2012.10.005
- Chouaibi, J., Rossi, M., & Abdessamed, N. (2021). The effect of corporate social responsibility practices on tax avoidance: An empirical study in the French context. Competitiveness Review, 32(3), 326-349. https://doi.org/10.1108/ CR-04-2021-0062
- Chulkov, D., & Wang, X. (2023). Corporate social responsibility and financial reporting quality: Evidence from US firms. Studies in Economics and Finance, 40(3), 445-466. https://doi.org/10.1108/SEF-09-2022-0462
- Cui, J., Jo, H., & Na, H. (2018). Does corporate social responsibility affect information asymmetry? Journal of Business Ethics, 148(3), 549-572. https://doi.org/10.1007/s10551-015-3003-8
- Darrag, M., & Crowther, D. (2017). Reflections on CSR: The case of Egypt. Society and Business Review, 12(1), 94-116. https://doi.org/10.1108/SBR-01-2016-0010
- Dechow, P. M., Kothari, S. P., & L. Watts, R. (1998). The relation between earnings and cash flows. Journal of Accounting and Economics, 25(2), 133-168. https://doi.org/10.1016/S0165-4101(98)00020-2
- Desoky, A. M., & Mousa, G. A. (2012). Corporate governance practices: Transparency and disclosure-evidence from the Egyptian exchange. Journal of Accounting, Finance and Economics, 2(1), 49–72.
- Diab, A., Abdelazim, S. I., & Aly, S. A. S. (2024). The moderating effect of audit quality and COVID-19 on the dividend payout-firm performance relationship: Egypt evidence. Cogent Business & Management, 11(1), 2297464. https://doi. org/10.1080/23311975.2023.2297464
- Diab, A., Abdelazim, S. I., & Metwally, A. B. M. (2023). The impact of institutional ownership on the value relevance of accounting information: Evidence from Egypt. Journal of Financial Reporting and Accounting, 21(3), 509-525. https://doi.org/10.1108/JFRA-05-2021-0130
- Diab, A., & Metwally, A. B. M. (2020). Institutional complexity and CSR practices: Evidence from a developing country. Journal of Accounting in Emerging Economies, 10(4), 655-680. https://doi.org/10.1108/JAEE-11-2019-0214
- Dietrich, D. (2007). Asset tangibility and capital allocation. Journal of Corporate Finance, 13(5), 995-1007. https://doi. org/10.1016/j.jcorpfin.2007.05.001
- Du, M., & Li, Y. (2023). Tax avoidance, CSR performance and financial impacts: Evidence from BRICS economies. International Journal of Emerging Markets. https://doi.org/10.1108/IJOEM-05-2022-0747
- Dyreng, S. D., Hanlon, M., & Maydew, E. L. (2008). Long-run corporate tax avoidance. The Accounting Review, 83(1), 61-82. https://doi.org/10.2308/accr.2008.83.1.61
- Dyreng, S. D., Hanlon, M., & Maydew, E. L. (2010). The effects of executives on corporate tax avoidance. The Accounting Review, 85(4), 1163-1189. https://doi.org/10.2308/accr.2010.85.4.1163
- El-Bassiouny, D., & Letmathe, P. (2018). The adoption of CSR practices in Egypt: Internal efficiency or external legitimation? Sustainability Accounting, Management and Policy Journal, 9(5), 642-665. https://doi.org/10.1108/SAMPJ-10-2017-0126
- Eliwa, Y., Haslam, J., & Abraham, S. (2016). The association between earnings quality and the cost of equity capital: Evidence from the UK. International Review of Financial Analysis, 48, 125–139. https://doi.org/10.1016/j.irfa.2016.09.012
- Elmghaamez, I. K., Nwachukwu, J., & Ntim, C. G. (2023), ESG disclosure and financial performance of multinational enterprises: The moderating effect of board standing committees. International Journal of Finance & Economics, 29(3), 3593-3638. https://doi.org/10.1002/ijfe.2846
- Fama, E. F., & Jensen, M. C. (1983). Separation of ownership and control. The Journal of Law and Economics, 26(2), 301–325. https://doi.org/10.1086/467037
- Flammer, C. (2015). Does corporate social responsibility lead to superior financial performance? A regression discontinuity approach. Management Science, 61(11), 2549-2568. https://doi.org/10.1287/mnsc.2014.2038
- Francis, J., LaFond, R., Olsson, P., & Schipper, K. (2005). The market pricing of accruals quality. Journal of Accounting and Economics, 39(2), 295-327. https://doi.org/10.1016/j.jacceco.2004.06.003
- Gavious, I., Livne, G., & Chen, E. (2022). Does tax avoidance increase or decrease when tax enforcement is stronger? Evidence using CSR heterogeneity perspective. International Review of Financial Analysis, 84, 102325. https://doi. org/10.1016/j.irfa.2022.102325
- Grougiou, V., Leventis, S., Dedoulis, E., & Owusu-Ansah, S. (2019). Corporate social responsibility and earnings management in U.S. banks. Accounting Forum, 38(3), 155-169. https://doi.org/10.1016/j.accfor.2014.05.003
- Hamrouni, A., Bouattour, M., Ben Farhat Toumi, N., & Boussaada, R. (2021). Corporate social responsibility disclosure and information asymmetry: Does boardroom attributes matter? Journal of Applied Accounting Research, 23(5), 897-920. https://doi.org/10.1108/JAAR-03-2021-0056
- Hoi, C. K., Wu, Q., & Zhang, H. (2013). Is corporate social responsibility (CSR) associated with tax avoidance? Evidence from irresponsible CSR activities. The Accounting Review, 88(6), 2025-2059. https://doi.org/10.2308/accr-50544
- Hope, O.-K., Thomas, W. B., & Vyas, D. (2013). Financial reporting quality of U.S. private and public firms. The Accounting Review, 88(5), 1715-1742. https://doi.org/10.2308/accr-50494
- Jiang, W., Zhang, C., & Si, C. (2022). The real effect of mandatory CSR disclosure: Evidence of corporate tax avoidance. Technological Forecasting and Social Change, 179, 121646. https://doi.org/10.1016/j.techfore.2022.121646
- Jouber, H. (2019a). Corporate social responsibility and earnings quality: Do institutional features matter? Journal of Global Responsibility, 11(1), 54–92. https://doi.org/10.1108/JGR-04-2019-0041
- Jouber, H. (2019b). How does CEO pay slice influence corporate social responsibility? U.S.-Canadian versus Spanish-French listed firms. Corporate Social Responsibility and Environmental Management, 26(2), 502-517. https://doi. org/10.1002/csr.1728



- Kacem, H., & Brahim Omri, M. A. (2021). Corporate social responsibility (CSR) and tax incentives: The case of Tunisian companies. Journal of Financial Reporting and Accounting, 20(3/4), 639-666. https://doi.org/10.1108/JFRA-07-2020-0213 Kerr, J. N. (2013). The real effects of opacity: Evidence from tax avoidance.
- Kim, Y., Park, M. S., & Wier, B. (2012). Is earnings quality associated with corporate social responsibility?. The Accounting Review, 87(3), 761-796. http://www.jstor.org/stable/23245629
- Kılıç, M., & Kuzey, C. (2018). Determinants of forward-looking disclosures in integrated reporting. Managerial Auditing Journal, 33(1), 115-144. https://doi.org/10.1108/MAJ-12-2016-1498
- Kim, S., & Yoo, J. (2022). Corporate opacity, corporate social responsibility, and financial performance. Finance Research Letters, 49, 103118. https://doi.org/10.1016/j.frl.2022.103118
- Lastiati, A. (2020). The tax avoidance opacity [paper presentation]. 2nd ICAN Malaysia International Conference on Accounting and Finance (pp. 65-74). Kuching, Malaysia.
- Lee, C.-H., & Bose, S. (2021). Do family firms engage in less tax avoidance than non-family firms? The corporate opacity perspective. Journal of Contemporary Accounting & Economics, 17(2), 100263. https://doi.org/10.1016/j. jcae.2021.100263
- Lee, L. F. (2012). Incentives to inflate reported cash from operations using classification and timing. The Accounting Review, 87(1), 1-33. https://doi.org/10.2308/accr-10156
- Lin, Y.-E., Li, Y.-W., Cheng, T. Y., & Lam, K. (2021). Corporate social responsibility and investment efficiency: Does business strategy matter? International Review of Financial Analysis, 73, 101585. https://doi.org/10.1016/j.irfa.2020.101585
- Liu, H., & Lee, H.-A. (2019). The effect of corporate social responsibility on earnings management and tax avoidance in Chinese listed companies. International Journal of Accounting & Information Management, 27(4), 632-652. https:// doi.org/10.1108/IJAIM-08-2018-0095
- Marwat, J., Rajput, S. K. O., Dakhan, S. A., Kumari, S., & Ilyas, M. (2021). Tax avoidance as earning game player in emerging economies: Evidence from Pakistan. South Asian Journal of Business Studies, 12(2), 186-201. https://doi. org/10.1108/SAJBS-10-2020-0379
- Metwally, A. (2022). The materiality of corporate governance report disclosures: Investigating the perceptions of external auditors working in Egypt. Scientific Journal for Financial and Commercial Studies and Researches (SJFCSR), 3(1 Part 1), 547-582. https://doi.org/10.21608/cfdj.2021.207396
- Metwally, A., Mohamed, A., & Ali, S. (2021). The management report usefulness in lending decision: Investigating Egyptian credit managers perceptions. Scientific Journal for Financial and Commercial Studies and Researches (SJFCSR), 2(1), 1-53. https://doi.org/10.21608/cfdj.2021.129326
- Metwally, A. B. M., Diab, A., & Mohamed, M. K. (2022). Telework operationalization through internal CSR, governmentality and accountability during the Covid-19: Evidence from a developing country. International Journal of Organizational Analysis, 30(6), 1441-1464. https://doi.org/10.1108/IJOA-11-2020-2500
- Mohamed, M. K., Allini, A., Ferri, L., & Zampella, A. (2019). Investors' perception on the usefulness of management report disclosures. Meditari Accountancy Research, 27(6), 893-920. https://doi.org/10.1108/MEDAR-11-2018-0393
- Mohmed, A., Flynn, A., & Grey, C. (2019). The link between CSR and earnings quality: Evidence from Egypt. Journal of Accounting in Emerging Economies, 10(1), 1-20. https://doi.org/10.1108/JAEE-10-2018-0109
- Muttakin, M. B., & Subramaniam, N. (2015). Firm ownership and board characteristics. Sustainability Accounting, Management and Policy Journal, 6(2), 138-165. https://doi.org/10.1108/SAMPJ-10-2013-0042
- Nair, R., Muttakin, M., Khan, A., Subramaniam, N., & Somanath, V. S. (2019). Corporate social responsibility disclosure and financial transparency: Evidence from India. Pacific-Basin Finance Journal, 56, 330-351. https://doi.org/10.1016/i. pacfin.2019.06.015
- Nguyen, L.-T. X. (2022). The relationship between corporate sustainability performance and earnings management: Evidence from emerging East Asian economies. Journal of Financial Reporting and Accounting, 22(3), 564-582. https://doi.org/10.1108/JFRA-09-2021-0302
- Ntim, C. G., & Soobaroyen, T. (2013). Corporate governance and performance in socially responsible corporations: New empirical insights from a neo-institutional framework. Corporate Governance: An International Review, 21(5), 468-494. https://doi.org/10.1111/corg.12026
- Prior, D., Surroca, J., & Tribó, J. A. (2008). Are socially responsible managers really ethical? Exploring the relationship between earnings management and corporate social responsibility. Corporate Governance: An International Review, 16(3), 160–177. https://doi.org/10.1111/j.1467-8683.2008.00678.x
- Rahman, M. J., & Zheng, X. (2022). Whether family ownership affects the relationship between CSR and EM: Evidence from Chinese listed firms. Journal of Family Business Management, 13(2), 373-386. https://doi.org/10.1108/JFBM-03-2022-0030
- Rakia, R., Kachouri, M., & Jarboui, A. (2023). The moderating effect of women directors on the relationship between corporate social responsibility and corporate tax avoidance? Evidence from Malaysia. Journal of Accounting in Emerging Economies. https://doi.org/10.1108/JAEE-01-2021-0029
- Richardson, G., Taylor, G., & Lanis, R. (2013). The impact of board of director oversight characteristics on corporate tax aggressiveness: An empirical analysis. Journal of Accounting and Public Policy, 32(3), 68-88. https://doi. org/10.1016/j.jaccpubpol.2013.02.004
- Sá, C., & Alves, H. (2022). Tax avoidance and companies' opacity: A theoretical approach. In D. Procházka (Ed.), Regulation of finance and accounting (pp. 421–434). Springer International Publishing.
- Salewski, M., & Zülch, H. (2014). The Association between corporate social responsibility (CSR) and earnings qualityevidence from European blue chips, working paper. HHL - Leipzig Graduate School of Management.

- Sial, M. S., Chunmei, Z., Khan, T., & Nguyen, V. K. (2018). Corporate social responsibility, firm performance and the moderating effect of earnings management in Chinese firms. Asia-Pacific Journal of Business Administration, 10(2/3), 184-199. https://doi.org/10.1108/APJBA-03-2018-0051
- UNCTAD. (2017). World investment report 2017. https://unctad.org/system/files/official-document/wir2017_en.pdf Van, H. V., & Ly, K. C. (2021). Does rising corporate social responsibility promote firm tax payments? New perspectives from a quantile approach. International Review of Financial Analysis, 77, 101857. https://doi.org/10.1016/j. irfa.2021.101857
- Wang, X. (2010). Tax avoidance, corporate transparency, and firm value [PhD dissertation]. University of Texas. World Bank. (2017). Doing business 2017 report. https://archive.doingbusiness.org/content/dam/doingBusiness/ media/Annual-Reports/English/DB17-Report.pdf
- You, L., & Chen, Z. (2022). A theory of firm opacity and corporate social responsibility. Journal of Banking & Finance, 145, 106640. https://doi.org/10.1016/j.jbankfin.2022.106640
- Zeng, T. (2019). Relationship between corporate social responsibility and tax avoidance: International evidence. Social Responsibility Journal, 15(2), 244–257. https://doi.org/10.1108/SRJ-03-2018-0056